Title: Comparison of the efficacy and safety of two starting dosages of prednisolone in early active rheumatoid arthritis (CORRA): study protocol for a randomized controlled trial

Authors:

Ulrike S Trampisch (Ulrike.Trampisch@uni-wh.de)
Dietmar Krause (krause@amib.ruhr-uni-bochum.de)
Hans J Trampisch (trampisch@amib.ruhr-uni-bochum.de)
Renate Klaassen-Mielke (klaassen-mielke.de@amib.ruhr-uni-bochum.de)
Xenofon Baraliakos (baraliakos@me.com)
Jürgen Braun (j.braun@rheumazentrum-ruhrgebiet.de)

Version: 5 Date: 11 August 2014

Author's response to reviews: see over
Dear Mrs. Lassere,

Many thanks for the reviewer’s recommendation and the editorial request. We respond point-by-point to the reviewer’s concerns.

1. The proposed sentence was added.

2. The full names of all ethical bodies that approved our study in the various centres involved were included.

Sincerely,

Dietmar Krause